

*J. Linguistics* **53** (2017), 463–499. © Cambridge University Press 2017 doi:10.1017/S002222671700010X First published online 15 May 2017

# Non-canonical agreement in copular clauses<sup>1</sup>

## SUSANA BÉJAR

University of Toronto

### ARSALAN KAHNEMUYIPOUR

University of Toronto Mississauga

## (Received 8 May 2014; revised 13 March 2017)

In this paper we investigate cross-linguistic variation in the morphosyntax of copular clauses, focusing on agreement patterns in binominal structures [NP1 BE NP2]. Our starting point is the alternation between NP1 and NP2 agreement, which arises both within and across languages. This alternation is typically taken to be confined to specificational (i.e. inverted) clauses, and previous analyses have strongly identified NP2 agreement with the syntax of inversion. However, we show that NP2 agreement is attested in a broader range of contexts, specifically in (assumed identity) equative structures, suggesting that it should not be correlated with specificational syntax. We present contrasting data from two languages - Persian and Eastern Armenian - for which the syntax of copular clauses is understudied. Whereas in Persian we see NP2 agreement in specificational structures but NP1 agreement in assumed identity equatives, in Eastern Armenian both types of structure yield NP2 agreement. We argue that the contrast between Persian and Eastern Armenian supports an approach that takes the NP1-NP2 alternation to arise as a phi-sensitivity in the probe-goal mechanics of Agree in a minimalist framework. Under this view, NP2 agreement is independent of syntactic inversion and is the result of the probe structure being articulated in such a way that certain NPs fail to Agree.

KEYWORDS: agreement, Armenian, copular clauses, equatives, specificational clauses

<sup>[1]</sup> We gratefully acknowledge funding from SSHRC grant # 410-2011-0975 and SSHRC grant #435-2013-1756. We would like to thank Caroline Heycock and Jutta Hartman for feedback and discussion, and four anonymous *Journal of Linguistics* referees for strengthening this work. We also thank present and past members of the Copular Agreement Project at the University of Toronto, including our co-investigator Ivona Kucerova and RAs Jitka Bartosova, Bronwyn Bjorkman, Cassandra Chapman, Julie Doner, Clarissa Forbes, Monica Irimia, Jessica Mathie, Kenji Oda, Will Oxford, Julia Su, Nicholas Welch, and Tomo Yokoyama. We also thank audiences at York University, the CLA, the Recife Agreement Workshop, the Bologna Workshop on Copular Clauses and the University of Toronto Syntax-Semantics Project. Finally, we are very grateful to the Persian and Eastern Armenian speakers who worked with us: Reza Savami (Persian), Ashot Karapet and John Nalbandyan (Eastern Armenian). Special thanks to Karine Megerdoomian for supplying Eastern Armenian judgments as well as additional help with transliteration.

Abbreviations used in the text are as follows: 1, 2, 3 = 1st, 2nd, 3rd person; ACC = accusative; EZ = Ezafe; FUT = future; GEN = genitive; INDEF = indefinite; LOC = locative; NEG = negation; NOM = nominative; PL = plural; PRES = present; PST = past; SG = singular; SP = specific; SUP = superlative. We capitalize 'Agree' when referring to the syntactic operation, but not when referring to the phenomenon.

#### 1. INTRODUCTION

The past twenty-five years have seen growing interest in non-canonical clausal agreement, i.e. a set of phenomena where the morphological cues signaling the relation between a predicate and one of its dependents fail to align in the expected manner.<sup>2</sup> However, these advances are skewed in that they are overwhelmingly focused on agreement in clauses with verbal predicates. This paper contributes to this vital area by investigating an understudied set of non-canonical agreement contexts found in copular clauses, i.e. sentences with non-verbal predicates. In particular, we examine binominal copular clauses, i.e. sentences where the copula links two nominals, as in [NP1 BE NP2].<sup>3</sup>

A striking peculiarity of binominal copular clauses is that there is variation across languages with respect to which noun phrase controls agreement. In some languages, for example English and French, agreement is consistently with the first noun phrase (NP1), while in others, such as German, Italian, Persian and Portuguese, a subset of copular clauses show agreement with the second noun phrase (NP2). We will refer to the latter pattern of agreement as an NP2 agreement pattern. This variation is illustrated in (1) with English, Portuguese (Costa 2004) and German.

(1)	(a)	The murderer is me.		(English; NP1 agreement)		
	(b)	Assassino sou eu. murderer am I		(Portuguese; NP2 agreement)		
	(c)	Der Mörder bist the murderer be.2SG	du you.NOM	(German; NP2 agreement)		

While copular constructions have long been an object of inquiry, the peculiarities of the NP2 agreement pattern – in particular person agreement with NP2 – and the formal derivational conditions that enable it have received relatively little attention, with the exception of groundbreaking work by Moro (1997) for Italian and subsequent work on Germanic languages (Den Dikken 1998, Heycock 2012) and Portuguese (Costa 2004).<sup>4</sup> Previous research in Generative syntax has converged on the idea that NP2 agreement follows from the syntax of a special class of copular clauses – specificational clauses – which have been argued to involve inversion (see Section 2 below).<sup>5</sup> This has been pursued along two lines:

<sup>[2]</sup> See, among many others, Barlow & Ferguson 1988, Laka 1993, Taraldsen 1995, Sigurðsson 1996, Hale 2001, Comrie 2003, Polinsky 2003 and Bobaljik & Branigan 2006.

<sup>[3]</sup> Following Moro (1988, 2013) we are using NP1 and NP2 as convenient labels for the two nominals. The numbers are intended to track surface order, which may be different from the base order. Note that the label NP is not intended as a commitment to the structure of the nominal. We use it for convenience, but will switch to DP when necessary, e.g. in Section 7.

<sup>[4]</sup> We set aside here the extensive literature on agreement in existential constructions which Moro 1997 relates to the syntax of specificational copular clauses.

<sup>[5]</sup> Moro (1997) further identifies NP2 agreement with the syntax of pro-drop languages. While Persian and Eastern Armenian are both pro-drop languages, this is not a hypothesis that we

- (i) NP1 has moved over NP2 to a position in the left periphery, as is shown by (2a) below, but NP2 is the actual structural subject and so NP2 agreement is in fact canonical agreement (Heggie 1988);
- (ii) NP1 has moved over NP2 into the structural subject position, as is shown by (2b). NP2 is not the structural subject, but at the point in the derivation when agreement occurs it is the closest available NP (Mikkelsen 2005, Heycock 2012, building on Moro 1997).



In this paper we assess two previously undescribed cases of NP2 agreement in Persian and Eastern Armenian.<sup>6</sup> These cases are distinct in that one (Persian) is restricted to specificational contexts and the other (Eastern Armenian) is seen in both specificational contexts and with equatives, the latter being especially noteworthy in light of claims in the previous literature linking NP2 agreement with specificational contexts. We develop an alternative account of NP2 agreement that dissociates it from the syntax of inversion. This account is framed within the minimalist theory of Agree (Chomsky 2000 inter alia), in which agreement is modeled as a formal relation between the agreement target (the probe) and the agreement controller (the goal). In this approach, probe-goal mechanics determine the availability of agreement. These include an operation that matches the probe and the goal (Match) and an operation that copies features of the goal to the probe (Agree). We propose that NP2 agreement arises when the phi-feature structure of NP1 is defective relative to the probe, with specificational contexts being just a subset of the relevant cases. This proposal permits a unification of our two case studies that would be challenging for previous accounts.

explore further in this paper, as there are important counterexamples to the generalization that NP2 agreement is related to *pro*-drop. Den Dikken (1998) gives the example of Dutch, a non-*pro*-drop language that has NP2 agreement. Other non-*pro*-drop languages for which NP2 agreement has been described in the literature include Faroese and German (Heycock 2012). The dissociation between *pro*-drop languages and NP2 agreement is further supported by the existence of NP1 agreement in Arabic, a *pro*-drop language.

<sup>[6]</sup> Persian is an Indo-European language spoken in Iran as well as some other countries in the region, such as Afghanistan and Tajikistan. The data in this paper is based on the dialect spoken in Iran's capital, Tehran. Armenian, also an Indo-European language, is spoken in Armenia, Iran and the diaspora. The data in this paper represent the Eastern dialect spoken in Iran and Armenia. Both Persian and Eastern Armenian are SOV.

The rest of this paper proceeds as follows. Section 2 gives background on the typology of copular clauses. Section 3 introduces the case of Persian NP2 agreement in specificational contexts. Section 4 establishes for Persian that NP2 is not the structural subject in the relevant cases. In other words, it is not the case that NP2 is the real subject and NP1 is simply fronted to a higher position (ruling out type (i) analyses). Section 5 introduces the Eastern Armenian case study. Crucially, we show that, in Eastern Armenian, NP2 agreement is not restricted to specificational contexts, but rather involves a sensitivity to 1st or 2nd person (participant) NP2s (ruling out type (ii) analyses). Section 6 provides an analysis of Eastern Armenian that situates NP2 agreement within the broader phenomenon of person-sensitivity. We extend this analysis to the Persian pattern in Section 7, thereby unifying both cases of NP2 agreement as resulting from a phi-defective NP1. Section 8 deals with contexts with two 3rd person NPs. Section 9 is a conclusion.

#### 2. COPULAR CLAUSES: SOME BACKGROUND

The syntax and semantics of copular clauses, while superficially simple, belies an underlying complexity and heterogeneity which have been intensively studied.<sup>7</sup> Since Higgins (1973), it is common to differentiate at least three subtypes of copular clauses: predicational, specificational, and equative. The examples in (3) illustrate the three major types of copular clauses:

(3)	(a)	Spiderman is my favourite superhero.	(predicational)
	(b)	My favourite superhero is Spiderman.	(specificational)
	(c)	Spiderman is Peter Parker.	(equative)

In (3a) NP1 is a referential noun phrase and NP2 is a nominal predicate. In (3b) it is NP2 that is referential, and NP1 introduces a description that is satisfied by NP2.<sup>8</sup> In (3c) both NP1 and NP2 can be construed as referential.

Higgins (1973) also introduced a fourth type referred to as identificational and shown in (4) below. Following Mikkelsen (2005) and Moltmann (2010) (contra Heller & Wolter 2008), we take identificational clauses to be a subtype of specificational clauses.

(4) This is Spiderman.

(identificational)

 <sup>[7]</sup> See, for example, among many others, Higgins 1973, 1979; Rothstein 1983, 1995, 2001;
 Rapoport 1987; Declerck 1988; Doron 1988; Heggie 1988; Heycock 1994, 2012; Moro 1997, 2006; Stassen 1997; Heycock & Kroch 1999; Dixon 2002; Adger & Ramchand 2003; Geist 2003; Pustet 2003; Heller 2005; Mikkelsen 2005; Den Dikken 2006a, b; Pereltsvaig 2007.

<sup>[8]</sup> The status of NP1 in specificational clauses has been subject to a great deal of debate. One approach (Mikkelsen 2005, following Partee 1986) has been to characterize NP1 as a predicate. This dovetails easily with the idea that the specificational type is simply an inversion of the predicational type. However, the identification of NP1 in specificational clauses as a predicate has been shown to be problematic (see Heycock & Kroch 1998, 1999; Heycock 2012). The characterization of NP1 in specificational clauses requires further elaboration and we will return to it in Section 7.

We now lay out our assumptions about the structure of the three major types of copular clauses. It is well established that the different types manifest special syntactic properties, e.g. connectivity effects, extraction patterns and information structure patterns (Higgins 1973, Rapoport 1987, Heggie 1988, Moro 1997, Heycock & Kroch 1999, Sharvit 1999, Mikkelsen 2005, Den Dikken 2006a, b). These have been argued to follow from distinct underlying structures and/or distinct syntactic derivations, with considerable heterogeneity among the existing analyses. The clearest syntactic and semantic differences separate specificational (i.e. inverted) clauses from the others. This is perhaps why NP2 agreement has been taken to follow from specificational syntax, one more in a cluster of its special properties. As stated above, we argue in this paper that NP2 agreement is not a reflex of inversion, but this does not undermine the inversion analysis and its ability to account for other distinctive properties.

While inversion is widely accepted as a crucial step in the derivation of specificational clauses, there are numerous unresolved questions. One prominent issue has to do with whether inverted (specificational) clauses take equative or predicational structures as their source. According to Heycock & Kroch (1998, 1999), Heycock (2012) and Den Dikken (2006a, b), specificational clauses and equatives share the same source structure, while for Moro (1997) and Mikkelsen (2005) the shared source is a predicational clause. Another related question is whether equatives have a distinct status. This question comes down to whether there is such a thing as a syntactically instantiated identity relation (as opposed to the more canonical predication relation). Some authors (Moro 1997, Adger & Ramchand 2003) reject the existence of an identity relation and argue that the underlying relation is predicational for all copular clause types. Others (Heycock & Kroch 1998, 1999; Mikkelsen 2005; Heycock 2012) propose that equatives are a distinct syntactic type, though there are differences of opinion with respect to how equatives relate to the other types (for Mikkelsen they are unrelated, for Heycock & Kroch (1998, 1999) and Heycock (2012), specificational clauses are a special case of equatives).

Abstracting away from the details reviewed above, all of these analyses reduce the syntax of copular clauses to two broad types – canonical vs. inverse order – and align the inverse order with specificational clauses. We assume the canonical structure in (5) below for predicational and equative clauses (following Moro 1997, Heycock & Kroch 1998, Mikkelsen 2005, Heycock 2012), and we assume the inverse structure in (6) for specificational clauses. (This structure is most closely aligned with Heycock 2012, but see also Moro 1997, Heycock & Kroch 1998, Mikkelsen 2005, Den Dikken 2006a, b).<sup>9</sup>

<sup>[9]</sup> Note that Den Dikken 2006 argues that equatives also involve inverse structures. See footnote 22 below on this point.

#### SUSANA BÉJAR & ARSALAN KAHNEMUYIPOUR

(5)	(a) (b)	T [NP1 NP2] NP1 T [t NP2]	(canonical copular clause)
(6)	(a)	T F [NP2 NP1	] (inverse copular clause)
	(b)	T NP1 F [NP2 $t$ ]	
	(c)	NP1 T $t$ F [NP2 $t$ ]	

In both (5) and (6), NP1 and NP2 are introduced in a small clause. In (6) the small clause is selected by a functional head F, which is the locus of the inversion step. We assume that subsequent movement of NP1 to the specifier of T occurs for reasons related to the EPP.

In the following section we turn to the Persian case study, where we will see that NP2 agreement aligns with the class of specificational clauses, as per previous claims in the literature.

#### 3. AGREEMENT IN PERSIAN COPULAR CLAUSES: BASIC FACTS

In this section, we lay out the agreement pattern in Persian copular clauses with two noun phrases, and show that Persian has NP2 agreement in specificational clauses but NP1 agreement elsewhere.

We begin by pointing out that Persian exhibits a straightforward pattern of subject agreement in core transitive and intransitive clauses, as shown in (7), with agreement highlighted in bold.

- (7) (a) (man) ye ketaab xarid-**am** I a book bought-1SG 'I bought a book.'
  - (b) Ali o Maryam man-o na-did-an Ali and Maryam I-ACC NEG-saw-3PL
     'Ali and Maryam didn't see me.'
  - (c) (man) to-ro dust daar-**am** I you(SG)-ACC friend have-1SG 'I love you.'
  - (d) naama-sh hamin ruz-aa mi-res-e
     letter-his/her these day-PL DUR-arrive-3SG
     'His/Her letter will arrive one of these days.'

The form of the verb co-varies with the person and number features of the subject. This is consistent with other nominative-accusative languages (e.g. English), where agreement correlates with nominative case assignment: the subject receives

nominative case and controls agreement on the verb. Agreement is never with the object noun phrase in these sentences.<sup>10</sup>

While agreement in non-copular contexts is consistently with the subject, agreement in the copular context is sometimes with NP1, as seen in the predicational sentences in (8), and sometimes with NP2, as in the specificational clauses in (9).<sup>11</sup>

- (8) NP1 agreement in Persian (predicational) copular clauses
  - (a) man barande-ye mosaabeqe (hast)-am
     I winner-EZ game be-1SG
     'I am the winner of the game.'
  - (b) to doktor-Ø-i you doctor-be-2SG 'You are a doctor.'
  - (c) maa behtarin daaneshaamuz-aan-e kelaas bud-im we best student-PL-EZ class be.PST-1PL
     'We were the best students of the class.'
- (9) NP2 agreement in Persian (specificational) copular clauses
  - (a) behtarin dust-e Soroosh man-Ø-am
     best friend-EZ Soroosh I-be-1SG
     'Soroosh's best friend is me.'
  - (b) qaatel to-Ø-yi murderer you.SG-be-2SG 'The murderer is you.'

<sup>[10]</sup> Persian is a null subject language (pro-drop), hence the optionality of overt pronominal subjects in (7). In (7c) and throughout the paper we are glossing -raa as an accusative marker for convenience and we are not taking a stance on the correct analysis of this marker of specific objects, nor do we take a stance on its status as an affix or clitic. Note that -raa sometimes appears as -ro or -o (after consonants) in colloquial Persian.

<sup>[11]</sup> Some notes on the data may be helpful. First, Persian has an overt copula *hastan* 'be' which is often deleted in colloquial speech. The presence or absence of the copula has no bearing on the main points discussed in this paper. Second, the form of the agreement suffix shows variation due to the presence of an epenthetic glide in some vowel hiatus contexts. Strikingly, we see in examples like (9), where the copula is deleted, that the agreement marker persists. This raises questions about the status of what has been traditionally regarded as an agreement suffix, suggesting the possibility that it is perhaps better analyzed as a clitic. We abstract away from this issue here and continue to refer to it as agreement, though its clitic-like properties become relevant in Section 4. Third, the Ezafe (glossed as EZ in this paper) is an unstressed vowel -e (-ye after consonants) which appears between a noun and its post-nominal modifier. If there is more than one modifier, the Ezafe vowel appears on all modifiers but the last one (see Kahnemuyipour 2014 and references cited therein).

#### SUSANA BÉJAR & ARSALAN KAHNEMUYIPOUR

(c) mohem-tarin sherkatkonande-haa-ye konferaans-e fardaa important-SUP participant-PL-EZ conference-EZ tomorrow shoamaa-Ø-yin you.PL-be-2PL
 'The most important participants of tomorrow's conference are you(PL).'

Note that to identify a clause as specificational special attention must be paid to its interpretation. Informally speaking, in a specificational interpretation NP1 introduces something like a description and NP2 specifies the entity or entities that fit the description. Furthermore, NP2 is typically new information for the hearer. Let us elaborate on the specificational reading of (9b). A felicitous context for this example with the intended reading would be one where a detective, let's say Hercule Poirot, has concluded a murder investigation and identifies the murderer by announcing: *The murderer is you*. We will refer to this as the Poirot reading.<sup>12</sup>

The Persian NP2 agreement pattern in (9) illustrates a clear departure from some other nominative-accusative languages such as English and French in which the agreement in copular sentences is no different from that in core transitive/intransitive sentences, i.e. agreement in copular clauses in these languages is consistently with NP1 as in (1a).

Persian identificational clauses, which we take to be a sub-type of specificational clause, also exhibit NP2 agreement, as shown in (10). For these examples a natural context would be identifying people in pictures.

(10)	(a)	in man-Ø- <b>am</b>	(d)	in maa-Ø <b>-yim</b>
		this I-be-1SG		this we-be-1PL
		'This is me.'		'This is us.'
	(b)	in to-Ø <b>-yi</b>	(e)	in shomaa-Ø <b>-yin</b>
		this you.SG-be-2SG		this you.PL-be-2PI
		'This is you.'		'This is you(PL).'
	(c)	in un-Ø-e	(f)	in unaa-Ø- <b>n</b>
		this him/her-be-3sG		this they-be-3PL

Unlike specificational/identificational clauses, equatives in Persian exhibit NP1 agreement. This is most apparent in assumed identity contexts like (11), where we can clearly contrast the phi-features of NP1 and NP2 (see Heycock 2012).

'This is them.'

'This is him/her.'

<sup>[12]</sup> We borrow the 'Poirot' context (as well as the 'dream' context used in Section 5) from Heycock (2010).

NON-CANONICAL AGREEMENT IN COPULAR CLAUSES

(11)	(a)	man to-Ø <b>-am</b>	(b)	to	man-Ø-i
		I you.sg-be-1sg		you	I-be-2sG
		'I am you.'		'You	ı are me.'

Assumed identity cases are felicitous in such contexts as dreams, or any other setting (e.g. the game of charades) where individuals swap identities. The examples in (11) are taken from a popular romantic song.

The contrast in agreement between equative and specificational clauses can be best seen in minimal pairs where a difference between NP1 and NP2 agreement forces an interpretive difference: NP1 agreement leads to an assumedidentity equative reading, while NP2 agreement requires a specificational reading. Heycock (2010) demonstrates this for German (12) and we can replicate the contrast in Persian (13). In (13a) the sentence has specificational semantics, whereas in (13b) we have an assumed identity interpretation.

(12) (a) Der Mörder **bist** du. (NP2 agreement) the murderer are.2SG you.NOM 'The murderer is you.'

[CONTEXT (Poirot reading): Poirot has concluded his investigation and identifies the murderer by announcing: The murderer is you.]

(b) Der Mörder **ist** du. (NP1 agreement) the murderer is.3SG you.NOM 'The murderer is you.'

[CONTEXT (charades reading): In a prison games room, a murderer, a thief and an arsonist are playing charades using themselves as characters. The arsonist says to the thief: The murderer is you.]

- (13) (a) qaatel to-Ø-yi murderer you-be-2SG 'The murderer is you.'
  - (b) qaatel to-Ø-e murderer you-be-3SG 'The murderer is you.'

Further examples of the same split are given in (14), where the specificational sentence (a) could be used in a context where the speaker is identifying herself ('Sabah' as a name), and the assumed identity sentence (b) could be used in a game of charades ('Sabah' as an individual).

- (14) (a) Sabah man-Ø-**am** Sabah I-be-1SG 'Sabah is me.'
  - (b) Sabah man-Ø-e Sabah I-be-3SG 'Sabah is me.'

This section has laid out the basic agreement facts in Persian copular clauses, where agreement with NP2 occurs in specificational (and identificational) clauses but not in predicational or equative clauses. We now turn our attention to the question of whether NP2 agreement in the specificational cases is truly a matter of non-canonical agreement. We examine an alternative possibility, that NP2 is the real subject and NP1 is fronted to a higher position, thus making the agreement underlyingly canonical. However, we reject this analytic possibility, a conclusion that is in keeping with other recent cross-linguistic analyses of specificational clauses.

#### 4. ON NP2 NOT BEING IN SPEC, TP IN PERSIAN

In this section we present several arguments against a topicalization analysis of NP1 which takes NP2 to be in subject position in specificational clauses, and we further show that these arguments can extend to languages with NP2 agreement. We then turn to Persian and provide several arguments to the same effect. We conclude that NP2 in Persian is consistently in a position below TP and is not the structural subject. For convenience, in what follows, we will describe this lower position as VP-internal, though given the dominant small clause analysis of copular clauses, VP could better be identified as PredP.

In seminal work by Heggie (1988), specificational clauses in English are analyzed as involving leftward movement of the copular complement (NP1) over the structural subject (NP2) to an A-bar position higher in the clause, as schematized in (2a) above, repeated here for convenience:



Under this approach, A-bar movement is taken to be an obligatory component of the syntax of specificational (inverted) clauses, but not predicational or equative (canonical) clauses. Such a structure allows for a straightforward analysis of NP2 agreement in specificational clauses as canonical subject agreement. However, subsequent work has converged on a different analysis for specificational clauses which still involves inversion but takes the surface position of NP1 to be the structural subject position of the clause, leaving NP2 lower in the clause (2b). Numerous arguments have been given to this effect for English.<sup>13</sup> For instance, Heycock (1994) and Rothstein (2001) argue that a Heggie-style analysis of specificational clauses (2a) is on the wrong track because it requires positing generalized movement of *be* to C (uncharacteristic of a non-V2 language) in order to derive the surface [NP1 BE NP2]. They also argue that (2a) makes

<sup>[13]</sup> We refer the interested reader to Mikkelsen (2005) and Bondaruk (2013) for reviews of the literature on this question.

incorrect predictions about word order in structures with modals (e.g. \**The winner might John be*). Further arguments are provided by Den Dikken (2006b) including subject–auxiliary inversion in (15), subject-to-subject raising in (16), binding in (17) and (18), and subordination in (19).

- (15) (a) The best candidate is Brian.
  - (b) Is the best candidate Brian?
- (16) The best candidate is likely \_ to be Brian.
- (17) (a) [Every country in Western Europe]<sub>i</sub> was the enemy of its<sub>i</sub> neighbour. (predicational)
  - (b) \*The enemy of its<sub>i</sub> neighbour was [every country in Western Europe]<sub>i</sub>. (specificational)

(Den Dikken 2006b: 155)

(18) (In the late 19th century Japan became a threat to its neighbours). Also a threat to its<sub>i</sub> neighbours was [every country/more than one country in Western Europe]<sub>i</sub>.

(Heycock & Kroch 1998: 81)

- (19) (a) It turned out that the applicant from New York was the best candidate. (predicational)
  - (b) It turned out that the best candidate was the applicant from New York. (specificational)

(Den Dikken 2006a: 353)

(20) (Taxes and inflation were always a problem for this government.) \*It turned out that also a problem were unemployment and labour unrest.

In (15) the specificational NP1 undergoes subject–auxiliary inversion; this is expected if NP1 is in Spec,TP but unexpected if NP1 is in a left-peripheral A-bar position. In (16) the specificational NP1 undergoes raising to subject, which is only expected if NP1 is in Spec,TP. In (17a) a pronoun in NP2 of a predicational clause can be bound by a quantifier in NP1, but in (17b), a specificational clause where by hypothesis the NP containing the pronoun has moved over the small clause subject, the pronoun cannot be bound. This is unexpected if inversion is to a left-peripheral A-bar position, on the assumption that A-bar movement feeds reconstruction. We can compare (17b) with a clear case of A-bar fronting (18), where binding of the fronted pronoun is possible. In (19b) we see that NP1 in specificational clauses is grammatical in subordinate contexts where A-bar fronting would be ungrammatical (20) (see also Heycock 1994):<sup>14</sup>

<sup>[14]</sup> Similar conclusions have been reached for other languages, e.g. German and Faroese (Heycock 2012), and Polish (Bondaruk 2013). In her detailed case study of specificational clauses in Danish, Mikkelsen (2005) likewise concludes that specificational clauses do not involve

Of course, English is an NP1 agreement language. One might argue that NP2 agreement languages should differ in precisely this respect, with NP2 in these languages being the structural subject of the clause. Below we present arguments against NP2 as structural subject in Persian, but before turning to Persian it should be noted that arguments against NP2 as structural subject in other NP2 agreement languages can be found in the literature, as well. For example, Den Dikken (2006a) shows that some of the same arguments used for English also apply to Dutch, an NP2 agreement language. In (21) we see that Dutch specificational clauses are grammatical in subordinate contexts, whereas Den Dikken notes that topicalization is impossible in embedded clauses in Dutch.

(21) (a) Het bleek dat de sollicitant uit NY de beste it turned.out that the applicant from NY the best kandidaat was. candidate was

'It turned out that the applicant from NY was the best candidate.'

 (b) Het bleek dat de beste kandidaat de sollicitant uit it turned.out that the best candidate the applicant from NY was.
 NY was

'It turned out that the best candidate was the applicant from NY.' (Den Dikken 2006a: 353)

In (22) we see that specificational subjects can undergo subject–auxiliary inversion. As before, this is unexpected unless NP1 is in subject position Spec,TP.<sup>15</sup>

- (22) (a) Je grootste zorg zijn de kinderen. your biggest worry is the children 'Your biggest worry is the children.'
  - (b) Zijn je grootste zorg de kinderen?is your biggest worry the children'Is your biggest worry the children?'

Having looked at several arguments against a topicalization analysis of specificational clauses in several NP1 and NP2 agreement languages, we now turn to Persian, for which we present arguments from prosody, cliticization patterns, adverb placement, VP-preposing and binding.

topicalization of the predicate to a position above the structural subject. She provides extensive diagnostics: position of negation, morphological case on NP2 when it is pronominal, absence of reconstruction effects, distribution of Negative Polarity Items, position of auxiliaries, availability of tag questions, and the possibility of embedding in contexts where predicate topicalization is impossible.

<sup>[15]</sup> The examples in (22) are adapted from Den Dikken (2010: 355), who reports (22b) to have two question marks. We have elicited judgments from four speakers. Three found (22b) to be fine, while one reported a slight degradation.

We begin by noting that the prosodic behaviour of copular clauses is suggestive of NP2 being in a VP-internal position. In Persian, the subject and the predicate form two separate domains with respect to sentential stress, with the leftmost element in the verb-phrase domain receiving primary stress and the subject receiving secondary stress (see Kahnemuyipour 2009). In all copular clauses, NP2 receives primary stress, which is consistent with a VP-internal position. An example is given in (23), where the full form of the verb 'be' is used to provide a stressable item in that position.

(23) behtarin dust-e Sabah <u>man</u> (hast)-am best friend-EZ Sabah I be.PRES-1SG 'Sabah's best friend is me.'

If the example in (23) was the result of true topicalization of NP1, one might argue that main stress should have been on the verb – see (24).<sup>16</sup> (Note: Underlining marks the word/phrase with the highest prosodic prominence at the sentence level.)

 (24) un futbaal-e mozaxraf-o man <u>did</u>-am that soccer-EZ terrible-ACC I saw-1SG
 'As for that terrible soccer game, I saw it.'

However, the argument from stress is undermined by a confound: specificational clauses across languages are characterized by a restrictive information structure such that NP2 is obligatorily focused. Focused nominals in Persian receive primary stress independent of their syntactic position (Kahnemuyipour 2009); thus, one could argue that the primary stress on NP2 in (23) above is a result of its being focalized and not a consequence of being in a low structural position. That said, this confound does not exclude the possibility that NP2 is indeed in a low focus position. See Kahnemuyipour 2001 for independent arguments in support of there being a low vP-internal focus position in Persian.

Stronger support for the idea that NP2 is in fact in a VP-internal position comes from the cliticization pattern in copular clauses where the copula can be deleted and its agreement marker cliticized to preceding VP-internal material, as noted earlier, in footnote 11. The agreement marker in (23) can cliticize to NP2 if the copula is null; this is another indication that NP2 is VP-internal. In fact, we can construct a minimal pair, as in (25), where both examples have a single NP but differ, under our analysis, as to whether that NP is VP-internal (corresponding to NP2 in a specificational clause with a null NP1) or VP-external.

<sup>[16]</sup> Primary stress on the subject in (24) would result in a contrastive reading.

- (25) (a) <u>man</u> (hast)-am I be.PRES-1SG 'It's me!'
  - (b) man \*(<u>hast</u>)-am I be.PRES-1SG 'I exist./I am here.'

In (25a), which has a presentational reading, we take the subject to be null and *man* 'I' to be the NP2 of an identificational clause (a sub-type of specificational clause per our assumptions). Note that cliticization to NP2 is possible here. In (25b), on the other hand, the interpretation we get is an existence reading for which we take *man* 'I' to be in subject position; this is further indicated by primary stress on the verb since it is at the left edge of VP. In this case, cliticization to the NP is impossible. One might wonder whether cliticization to the NP in (25b) is prohibited because – unlike (25a), where the NP is focused – here the NP is not focused, instead the copula is. However, (26) demonstrates that cliticization is bad in this structure even if the NP is focused.

- (26) A: Maryam o Ali \*(hast)-an? Maryam and Ali be-3PL 'Are Maryam and Ali here?'
  - B: <u>Pari o Taghi</u> \*(hast)-an, na Maryam o Ali <u>Pari and Taghi</u> be-3PL not Maryam and Ali 'Pari and Taghi are here, not Maryam and Ali.'

The dialogue in (26) sets up a contrastive focus reading for the subject of the second sentence (*Pari o Taghi*) but cliticization to this NP is not acceptable. Thus, we conclude that the cliticization contrast between (25a) and (25b) above does not reflect a difference in information structure but rather it reflects the different structural positions of the NP. Cliticization is only possible if, as per our analysis, the NP is VP-internal, as in (25a). In (27) we see another instance of the impossibility of cliticization to a VP-external NP, where the subject of a predicational clause cannot host the agreement clitic in the second conjunct, even when it is focused.

(27) to behtarin dust-e Soroosh hast-i yaa you best friend-EZ Soroosh be.PRES-2SG or man ??(hast)-am I be.PRES-1SG
 'Are you Soroosh's best friend or me?'

Another argument against the idea that NP2 is in Spec,TP in NP2 agreement cases comes from the position of temporal adverbs. While Persian typically allows a wide range of word orders, the most natural position for temporal adverbs is after the subject, as shown in (28) for a canonical transitive clause.

- (28) (a) man diruz ye futbaal-e mozaxraf did-am I yesterday a soccer-EZ terrible saw-1SG
   'I saw a terrible soccer game yesterday.'
  - (b) man emruz Hasan-o na-did-am I today Hasam-ACC NEG-saw-1SG 'I didn't see Hasan today.'

When it comes to copular sentences, the most natural place for the temporal adverb is after NP1, which is expected if we take NP1 to be in the subject position Spec,TP. We can see in (29a) that in a predicational copular clause, the temporal adverb goes after NP1.

- (29) (a) man emruz behtarin dust-e Soroosh-Ø-am
   I today best friend-EZ Soroosh-be-1SG
   'I am Soroosh's best friend today.'
  - (b) behtarin dust-e Soroosh emruz man-Ø-am best friend-EZ Soroosh today I-be-1SG
     'Soroosh's best friend is me today.'
  - (c) ??behtarin dust-e Soroosh man emruz hast-am
     best friend-Ez Soroosh I today be.PRES-1SG
     [Intended meaning: 'Soroosh's best friend is me today.']
  - (d) ketaab-o man diruz xarid-am book-ACC I yesterday bought-1SG 'The book, I bought it yesterday.'

The example in (29b) shows a specificational copular clause where the position of the temporal adverb is still after NP1 despite there being NP2 agreement. Crucially, if NP2 was in subject position in (29b), the temporal adverb should be able to follow it, contrary to fact (29c).<sup>17,18</sup> However, in true topicalization cases, the temporal adverb can follow the subject, as seen in (29d).

An anonymous JL referee notes that the degraded word order in (29c) is likewise observed in (30c), which is not a specificational clause but rather a predicational clause with a fronted adjectival predicate.

<sup>[17]</sup> In example (29c), the full form of the verb 'be' has been used to eliminate a confound that could otherwise arise because the agreement marker cannot cliticize to the adverb.

<sup>[18]</sup> An anonymous JL referee points out that (29c) is improved with a strong contrastive focus on the adverb coupled with marked prosodic prominence. It is worth noting that even with a contrastive reading of the adverb the most natural position would be clause-initial or after NP1. More importantly, introducing contrastive focus on the adverb involves a more marked construction. Our observation about adverb placement is crucially intended to apply to a specificational clause uttered in a neutral context, and there the most natural position is clearly after NP1.

- (30) (a) man naadaan bud-am I ignorant be.PST-1SG 'I was ignorant.'
  - (b) naadaan man bud-am (na to) ignorant I be.PST-1SG not you 'It was me who was ignorant (not you).'
  - (c) ??naadaan man saal-e pish bud-am (na to) ignorant I year-EZ previous be.PST-1SG not you
     'It was me who was ignorant last year (not you).'

Here, the sole NP is plausibly in subject position.<sup>19</sup> (30a) shows the corresponding predicational clause without fronting; (30b) shows the fronted version without the adverb. Since there is degradation in cases like (30c), where the low NP is plausibly in subject position, one might argue that this undermines the claim that the degradation in (29c) is because the lower NP is not a subject. In other words, the parallel between (29c) and (30c) might be taken to suggest that in both cases NP2 is in a subject position. However, we argue that the parallel between specificational clauses and copular clauses with fronted adjectival predicates is illusory. For one, while both (29b) and (30b) are similar with respect to information structure insofar as the low NP must be focalized, they differ in that (30b) only allows contrastive focus, whereas a specificational clause (29b) allows information focus as well. Thus, (29b) is felicitous as the answer to a context question like (31a) but (30b) is infelicitous as an answer to the parallel context question in (31b).<sup>20</sup>

- (31) (a) ki behtarin dust-e Soroosh-e who best friend-EZ Soroosh-be.3SG'Who is Soroosh's best friend?'
  - (b) ki naadaan bud who ignorant be.PST'Who was ignorant?'

A closer look at adverb placement patterns reinforces the conclusion that (29b, c) and (30b, c) do not involve parallel structures. While placing the adverb after the lower NP2 results in a degraded (29c), it is natural after NP1 in (29b). There is no equivalently natural position for the adverb in the predicational clause with a fronted adjectival predicate, as (32) shows.

<sup>[19]</sup> We do not aim to offer an analysis of the fronted adjectival clauses. While it seems plausible that the sole NP in these structures is in subject position, a closer examination of this construction might reveal otherwise. Cases of agreement with subjects in lower subject positions have been observed in other languages, for example Romance verb–subject structures (Rizzi 1982 inter alia).

<sup>[20]</sup> Note that both context questions can felicitously be answered by a canonical predicational clause with focus on the subject. For a predicational clause with an adjectival predicate, this is the only felicitous option.

(32) ?naadaan saal-e pish man bud-am (na to) ignorant year-EZ previous I be.PST-1SG not you
 'It was me who was ignorant last year (not you).'

We conclude that the naturalness of the position of the adverb after NP1 in (29b) is a good indication that NP1 actually is in the subject position.

The next argument against NP2 as subject comes from VP-preposing in Persian. Persian has a very productive VP-preposing operation which takes the predicate and places it before the subject, resulting in a sentence with a marked information structure (Kahnemuyipour 2009). This operation is shown for a non-copular sentence in (33).

- (33) (a) man ye ketaab-e xub xarid-am (canonical order) I a book-EZ good bought-1SG 'I bought a good book.'
  (b) ye ketaab-e xub xarid-am man (VP-preposed order) a book-EZ good bought-1SG I
  - 'I bought a good book.'

When this operation is applied to a specificational copular clause like (23) above, it takes NP2 with it, as shown in (34), which is the only acceptable way to topicalize the copula, providing support for the VP-internal position of NP2.

(34) man (hast)-am behtarin dust-e Sabah
I be.PRES-1SG best friend-EZ Sabah
'Sabah's best friend is me.'
[Information structure is not shown in the translation.]

It should be noted that in a scrambling language like Persian, word order permutations may arise through different mechanisms with varying information structure effects. For example, one might raise the possibility that an order like that in (34) above derives from a predicational clause where *man* 'I' is the subject and the predicate is postposed. However, we note that cliticization of the agreement marker to *man* is possible in (34). As discussed above, this is not expected to be possible if *man* is external to the VP as it would be if the source structure were a predicational clause. The preposing analysis of (34) is further supported by the distribution of the VP-adjoined adverb *qat'an* 'definitely', as in (35), where the adverb appears on the left edge with the preposed VP.

(35) qat'an man (hast)-am behtarin dust-e Sabah definitely I be.PRES-1SG best friend-EZ Sabah 'Sabah's best friend is definitely me.'
 [Information structure is not shown in the translation.]

If we compare the VP-preposed specificational structure in (34)–(35) with a postposed counterpart of (33a), shown in (36), we see a clear contrast: while the postposed (36a) is marked yet felicitous in an appropriate context, the addition of the adverb in (36b) is ungrammatical.

- (36) (a) ?man xarid-am ye ketaab-e xub (postposed order) I bought-1SG a book-EZ good 'I bought a good book.'
  - (b) \*?qat'an man xarid-am ye ketaab-e xub (postposed order) definitely I bought-1SG a book-EZ good
     'I definitely bought a good book.'

This suggests that the clause-initial position of the adverb is incompatible with postposed structures, providing further support for the idea that (34)–(35) are not postposed.

The final argument comes from binding. Recall from the discussion of English (17) above that a pronoun in NP1 of a specificational clause cannot be bound by a quantifier in NP2. This is unexpected if inversion is to a left-peripheral A-bar position, on the assumption that A-bar movement feeds reconstruction, which is supported for English by (18). These results are replicated in Persian. In a predicational clause (37), a pronoun in NP2 can be bound by a quantifier in NP1, but in a specificational clause (38) a pronoun in the inverted NP1 cannot be bound by a quantifier in NP2.

 (37) [har keshvar-i dar orupaay-e qarbi]<sub>i</sub> doshman-e every country-INDEF in Europe-EZ western enemy-EZ hamsaay(e)-ash<sub>i</sub> bud neighbour-its was

'[Every country in western Europe]<sub>i</sub> was the enemy of its<sub>i</sub> neighbor.'

(38) \*doshman-e hamsaay(e)-ashi [har keshvar-i dar enemy-EZ neighbour-its every country-INDEF in orupaay-e qarbi]i bud Europe-EZ western was
'The enemy of itsi neighbour was [every country in Western Europe]i.'

Meanwhile, (39) shows that with A-bar fronted VP-preposing, a pronoun in the fronted constituent can be bound by a quantifier to its right, suggesting that A-bar movement feeds reconstruction for binding purposes in Persian.

(39) doshman-e hamsaay(e)-ash<sub>i</sub> bud [har keshvar-i dar enemy-EZ neighbour-its was every country-INDEF in orupaay-e qarbi]<sub>i</sub>
Europe-EZ western
'[Every country in western Europe]<sub>i</sub> was the enemy of its<sub>i</sub> neighbor.'
[Information structure is not shown in the translation.]

We conclude from these facts that Persian specificational clauses do not involve A-bar fronting.

In this section we have provided arguments from the literature against a topicalization analysis of the inverted NP1 in specificational clauses and have added Persian-specific arguments from prosody, cliticization, adverb placement, VP-preposing and binding. On the basis of these arguments, we rule out an analysis of the NP2 agreement pattern as canonical subject agreement (contra (2a) above). In Section 6 we develop an alternative analysis of these agreement facts. In this analysis, NP1 is taken to be in the Spec,TP subject position and NP2 in a VP-internal position regardless of which NP controls agreement. We now turn to the Eastern Armenian case study, where we show that NP2 agreement is not limited to specificational clauses.

#### 5. NP2 AGREEMENT IN EASTERN ARMENIAN

We begin by laying out the basic facts of copular clause agreement in Eastern Armenian. Predicational copular clauses like (40) show NP1 agreement, and specificational clauses like (41) manifest NP2 agreement.

- (40) (a) yes arač<sup>h</sup>in hok<sup>h</sup>i-n ey aysteq I first soul-SP be.PST.1SG here 'I was the first person here.'
  - (b) du mart<sup>h</sup>aspan es you murderer be.PRES.2SG
     'You are a murderer.'
- (41) (a) arač<sup>h</sup>in mart<sup>h</sup>-ə aysteq yes ey first person-SP here I be.PST.1SG
   'The first person here was me.'
  - (b) mart<sup>h</sup>aspan-ə du es murderer-SP you be.PRES.2SG
     'The murderer is you.'

So far this pattern is reminiscent of Persian. Also as in Persian, we find evidence that NP2 in Eastern Armenian is low, and not in the structural subject position. The first piece of evidence comes from the position of temporal adverbs. In Eastern Armenian, the most natural place for the temporal adverb is after the subject, as we can see in the predicational clause in (42).

(42) yes (esor) Ashot-i lavaguyn ənker-ən (\*esor) em
 I today Ashot-GEN best friend-SP today be.PRES.1SG
 'I am Ashot's best friend today.'

Crucially, placing the temporal adverb after NP2 is ungrammatical in (42), indicating that the temporal adverb cannot appear internal to the VP domain. In the inverted (specificational) (43), the preferred position for the temporal adverb is still after NP1, and placing the adverb after NP2 is ungrammatical.

(43) Ashot-i lavaguyn ənker-ə (esor) yes (\*esor) em
 Ashot-GEN best friend-SP today I today be.PRES.1SG
 'Ashot's best friend is me today.'

This provides evidence against an analysis which takes NP2 to be in the structural subject position, and NP1 to be in a fronted A-bar position. Note that in a truly topicalized sentence, the temporal adverb can appear after the second NP (44).

(44) gIrk<sup>h</sup>-ə (yerek) yes (yerek) afel em
 book-SP yesterday I yesterday bought be.PRES.1SG
 'The book, I bought yesterday.'

Another piece of evidence against NP2 being the structural subject comes from VP preposing. Armenian has a productive process that fronts the VP, as in (45) (see Kahnemuyipour 2009).

(45) mi hat lav gırk<sup>h</sup> em aíel yes one CLASSIFIER good book be.1SG bought I
 'Buy a good book, I did.'

The process also applies to copular clauses, as in the specificational example in (46b) – derived from (46a) – where the copula fronts with NP2, suggesting that both are inside VP.

(46)	(a)	Soroosh-i	lavaguyn	ənker-ə	yes	em	
		Soroosh-SP	best	friend-SP	Ι	be.PR	es.1sg
		'Soroosh's l	best friend	is me.'			
	(b)	yes em	Sor	oosh-i	lava	aguyn	ənker-ə
		I be.PRE	s.1sg Sor	oosh-GEN	bes	t	friend-SP
		'It's me who is Soroosh's best friend.'					

There is a potential confound in (46b) because it is string ambiguous with a predicational clause in which NP1 has been focalized as in (47b), which would be derived from the predicational clause in (47a).

- (47) (a) yes Soroosh-i lavaguyn ənker-ə em I Soroosh-GEN best friend-SP be.PRES.1SG 'I am Soroosh's best friend.'
  (b) yes em Soroosh-i lavaguyn ənker-ə
  - I be.PRES.1SG Soroosh-GEN best friend-SP

'I (EMPH) am Soroosh's best friend.'

This is due to a special property of the Eastern Armenian copula (and auxiliary) which obligatorily cliticizes to focalized elements (Comrie 1984; Tamrazian 1994; Tragut 2009; Kahnemuyipour & Megerdoomian 2011, 2017). This confound is compounded by the fact that the pragmatic conditions under which (46b) and (47b) are felicitous are difficult to differentiate. To circumvent this problem, we show the preposing pattern using the future form  $k \geqslant lini$ , which does not have the aforementioned clitic properties. Thus (48b), the preposed counterpart of (48a), is not ambiguous, and we still see that NP2 fronts with the copula, suggesting that it is VP-internal.

- (48) (a) lavaguyn xaqac<sup>h</sup>oq-ə Ronaldo-n kə-lin-i best player-SP Ronaldo-SP FUT-be.SUBJUNCTIVE-3SG 'The best player will be Ronaldo.'
  - (b) [Ronaldo-n kə-lin-i] lavaguyn Ronaldo-SP FUT-be.SUBJUNCTIVE-3SG best xaqac<sup>h</sup>oq-ə player-SP
     'It will be Ronaldo (who is) the best player.'

So far, we have shown that Eastern Armenian greatly resembles Persian insofar as it has NP2 agreement in specificational clauses and NP2 is lower than the structural subject position. However, there is a crucial difference between Persian and Eastern Armenian, which we turn to next. In Persian assumed identity contexts, agreement is consistently with NP1 (see (13) and (14) above), whereas in Eastern Armenian, NP2 agreement can arise even in assumed identity contexts. The availability of NP2 agreement in assumed identity clauses is a counterexample to the claim that NP2 agreement is a consequence of specificational syntax (Sigurðsson 2006, Heycock 2012). In light of this we give multiple contexts in order to demonstrate the robustness of the NP2 agreement pattern. What we see is that NP2 agreement always arises when NP1 is 3rd person and NP2 is 1st or 2nd person, as in (49a, b), (50a, b) and (51a, b).

(49) Charades context

[CONTEXT: Shadi, Lina, Karine and Kamnoosh are playing a version of charades where instead of enacting names of movies, books, etc., everyone puts their name in a hat, and players must pantomime one another. The four have just finished their pantomimes. Karine, addressing Kamnoosh, says:]

- (a) Shadi-n yes ei/\*er
   Shadi-SP I be.PST.1SG/be.PST.3SG
   'Shadi was me.'
- (b) Lina-n du eir/\*er Lina-SP you be.PST.2SG/be.PST.3SG 'Lina was you.'

- (c) yes Shadi-n ei I Shadi-SP be.PST.1SG 'I was Shadi.'
- (d) du Lina-n eir you Lina-SP be.PST.2SG'You were Lina.'
- (50) Dream context

[CONTEXT (adapted from Heycock 2010): A prison counselor has just completed a session with several inmates (a murderer, a thief, etc.) where one of the inmates shared a crazy dream. Later, the counselor recounts this dream to a colleague:]

(a) ays yeraz-um, martaspan-ə yes ei isk goq-ə in dream-LOC murderer-SP I be.PST.1SG but thief-SP du eir you be.PST.2SG
 (b) the base of the set of the

'In the dream, the murderer was me but the thief was you.'

[CONTEXT: Kamnoosh and Karine are talking and Karine tells Kamnoosh about a dream she had involving their mutual friends Shadi and Lina.]

- (b) Shadi-n yes ei/\*er isk Lina-n du Shadi-SP I be.PST.1SG/3SG but Lina-SP you eir/\*er be.PST.2SG/3SG 'Shadi was me but Lina was you.'
  (c) yes Shadi-n ei
  - I Shadi-SP be.PST.1SG 'I was Shadi.'
- (d) du Shadi-n eir you Shadi-SP be.PST.2SG
   'You were Shadi.'
- (51) Comparison and contrast

[CONTEXT: Arsalan and Kamnoosh, in a conversation about their children, remark on how much the kids have come to resemble them.]

- (a) Soroosh-> (čisht) du es/\*e
   Soroosh-SP exact you be.PRES.2SG/be.PRES.3SG
   'Soroosh is (exactly) you.'
- (b) Sabah-n (čisht) yes em/\*e
   Sabah-SP exact I be.PRES.1SG/be.PRES.3SG
   'Sabah is (exactly) me.'

[CONTEXT: Shadi reacts to being compared unfavourably with a friend.]

(c) yes inqə č<sup>h</sup>-em, inqən-el yes I s/he NEG-be.PRES.1SG s/he-also I č<sup>h</sup>-??i/em NEG-be.PRES.3SG/1SG
'I am not her/him, s/he isn't me.'

[CONTEXT: A romantic song in which the singer identifies themself with their lover.]

(d) yes inqn/na em/\*e, inqə/na yes em/\*e I s/he be.PRES.1SG/3SG s/he I be.PRES.1SG/3SG 'I am her/him, s/he is me.'

If NP1 is 1st or 2nd person, then NP1 agreement arises, as we see in (49c, d), (50c, d) and (51c, d).<sup>21</sup> We take this pattern to indicate a preference for agreement with a 1st or 2nd person NP over a 3rd person NP.<sup>22</sup>

In this section we have seen that in Eastern Armenian agreement is with NP1 for predicational clauses, NP2 for specificational clauses, and NP1 or NP2 for equative (assumed identity) clauses, depending on the phi-features of the NPs. We will take this phi-sensitivity as our point of departure in the next section, where we formulate an analysis of the Eastern Armenian facts. We then turn to the Persian pattern and provide an account that unifies the two cases.

#### 6. NP2 AGREEMENT AS A PHI-SENSITIVITY

In this section we provide an analysis of the Eastern Armenian copular agreement pattern as a person-sensitivity.<sup>23</sup> The challenge is to capture the alternation between NP1 and NP2 agreement, as outlined above. Our starting point is the 1st/2nd person preference observed in the assumed identity (equative) clauses

<sup>[21]</sup> Speakers report certain unease with some of these examples, for example (49) and (51c). Crucially, even in these examples there is a clear preference for the indicated agreement patterns. In (49), an alternative construction using a genitive NP1 (possibly with an elided head, e.g. 'Shadi's [role]') is preferred.

<sup>[22]</sup> Den Dikken (2006b) has an analysis of equatives that involves an inversion step analogous to specificational clauses. This might seem to pave the way for a unification of all NP2 agreement cases with the syntax of inversion. However, this approach does not align with NP2 agreement any better than the more traditional view, which limits inversion to specificational clauses. The person-sensitive alternation between NP1 agreement and NP2 agreement in Eastern Armenian equatives cannot be explained on the basis of inversion alone. Nor can the contrast between Persian and Eastern Armenian with respect to NP2 agreement in equative clauses. Thus, for convenience, we adhere to the more conventional treatment of equatives as canonical.

<sup>[23]</sup> Agreement in Eastern Armenian is for both person and number, but as we see in Section 8 examples (54a, b), (55a, b) and (56a, b), number does not determine the choice of the controller of agreement, it merely follows parasitically. The crucial case is (56a), where a singular NP1 requires singular agreement even though a plural NP2 is available. This example shows that the probe is only sensitive to person features, which in this case are satisfied by the higher singular NP.

in (49)–(51). The generalization there was that NP2 agreement always arises when NP1 is 3rd person and NP2 is 1st or 2nd person. If NP1 is 1st or 2nd person, then agreement is with NP1. Below we give an analysis that captures these generalizations, and then we extend this analysis to specificational clauses in Eastern Armenian. We set aside copular clauses with two 3rd person NPs until Section 8.

We develop the analysis within Chomsky's (2000, 2001) Agree framework, where the operation Agree is defined as a relation between two elements in a syntactic domain: a probe and a goal. The probe is an unvalued set of features on a functional head that must receive a value from some other syntactic constituent, i.e. the goal. The unvalued probe triggers the search for a goal in its search domain (i.e. its c-command domain) and defines the search criteria (e.g.  $\phi$  probes  $\phi$ , WH probes WH, etc.). The operation whereby a goal satisfies the search criteria defined by the probe is called Match and the operation whereby the unvalued probe is valued by a matching goal is Agree. When more than one potential goal exists in the search domain, it is the closest matching element that Agrees, where closeness is defined by c-command.

Our first objective is to understand how NP2 agreement takes place in assumed identity (equative) clauses where NP1 is 3rd person and NP2 is 1st or 2nd person. First of all, if both NP1 and NP2 are potential controllers, then it must be the case that they share a single AGR domain. We can also deduce that the probe must be higher in the structure than NP1. We can establish this because there is a clear preference for NP1 agreement: if NP1 is 1st or 2nd person, then it will Agree, regardless of the features of NP2 – see (52) (see also Section 8). If the probe were located below NP1 we would expect NP2 agreement to be privileged, with NP1 agreement arising only when NP2 agreement failed (Béjar 2003).

- (52) [CONTEXT: Romantic declaration.]<sup>24</sup>
  - (a) yes du em/\*esI you be.PRES.1SG/be.PRES.2SG'I am you.'
  - (b) du yes \*em/es you I be.PRES.1SG/be.PRES.2SG 'You are me.'

We assume the probe is in T (head of the Tense Phrase TP). This is further suggested by the NOM–NOM case pattern in Eastern Armenian copular clauses.

<sup>[24]</sup> While the judgments in (52) are robust, other double participant cases that we have elicited in Eastern Armenian are not all as straightforward. It was generally easy to elicit cases with 1st person NP1 and 2nd person NP2 (with NP1 agreement). Cases of 2nd person NP1 and 1st person NP2 were harder to elicit in charades and dream contexts (though they still showed a preference for NP1 agreement).

This basic Agree configuration is schematized in (53), where the unvalued probe in preceded by an underscore:

(53)  $\dots T_{AGR \ \varphi} \dots NP1 \varphi \dots NP2 \varphi \dots$ 

While this configuration captures the preference for NP1 agreement and the shared AGR domain, it does not in itself explain how the NP1–NP2 alternation arises. We take the 1st/2nd person preference to suggest a person-hierarchy effect reminiscent of phenomena that have been observed in many other agreement systems (see Silverstein 1976, Zwicky 1977, Macaulay 1992, Hale 2001, Nichols 2001, Béjar 2003 and Béjar & Rezac 2009). We model the preference pattern using the general mechanics of Chomsky's Agree framework as described above, with modifications adapted from Béjar (2003) and Béjar & Rezac (2009), who build in person-sensitivities by manipulating the feature structure of the probe. This modifies the search criteria that determine whether NP1 or NP2 will count as the goal. In other words, the alternation between NP1 and NP2 agreement reduces to a locality effect. A priori, NP1 is privileged because it is the closest NP in the domain of the  $\phi$  probe. Only when NP1 fails to Agree (that is, when it fails to meet the criteria established by the probe) can an agreement relation be established with the farther NP2.

Whereas in the conventional implementation of phi-Agree (Chomsky 2000, 2001), the probe searches for phi-features as a bundle, the modified Agree mechanics that we adopt assume a more fine-grained probe that searches for particular phi-features. The refined search criteria needed to capture the Eastern Armenian facts require a feature system that treats 1st and 2nd persons as a natural class. We adopt the [participant] feature of Harley & Ritter (2002; see also Halle 1997, among others), which identifies a natural class of 1st and 2nd person, to the exclusion of 3rd person. The contrast between 1st and 2nd person is modeled, in turn, by a [speaker] feature on 1st person forms. We also assume that the minimal feature that all person forms have is a deictic feature [d]. The [d] feature thus provides a means of identifying all deictic NPs as a natural class. Note that the features are ordered by semantic entailment relations. This is reflected in the representations below, where lower features entail higher features.<sup>25</sup>

(54)	(a)	First person	(b)	Second person	(c)	Third person
		[d]		[d]		[d]
		[participant]		[participant]		
		[speaker]				

We take the probe in Eastern Armenian to be specified for the features [d] and [participant]. In this system, in order for Agree to succeed, the features of the goal

<sup>[25]</sup> Note that in this paper we refer to the [d] feature as a person feature and agreement with [d] as person agreement, as it stands in an entailment relation with [participant]. Also, in some systems there may be more feature structure between [d] and [part], in particular a minimal person feature [ $\pi$ ] that distinguishes animate/human 3rd persons. However, we abstract away from this here, as it does not appear to play a role in the patterns we are dealing with.

must be a superset of the probing features (see Béjar 2003, 2008 for details).<sup>26</sup> The phi-features of a 1st or 2nd person NP will be a superset of the probe and will Agree, but the phi-features of a 3rd person NP will not. Therefore, if NP1 is 1st or 2nd person, it halts the search making NP2 agreement impossible as in (55). This covers the cases of equative NP1 agreement that we saw in (49c, d), (50c, d), (51c, d), and (52) where NP1 is 1st or 2nd person. NP2 agreement becomes available when NP1 is 3rd person and fails to Match/Agree as in (56). This covers the cases of equative NP2 agreement that we saw in (49a, b), (50a, b) and (51a, b).<sup>27</sup>

$$(55) \longrightarrow (55) \longrightarrow (55) \longrightarrow (55) \longrightarrow (56) \longrightarrow$$

We should note that the same analysis can capture the cases of NP2 agreement with 1st or 2nd person NP2 that we saw in Eastern Armenian specificational clauses (41) above. This follows from the fact that NP1 is 3rd person and fails to Agree, similar to (56). We will, however, have more to say about the feature structure of NP1 in specificational clauses in Section 8, with technical implications for Eastern Armenian.

In this section we have laid out an analysis which takes the failure of NP1 agreement to depend on its status as 3rd person. This analysis captures the distribution of NP2 agreement in Eastern Armenian equatives and specificational clauses with 1st and 2nd person NPs. Still to be addressed are equative and specificational clauses with no 1st or 2nd person NP, as well as predicational clauses. We will return to these in Section 8. But first we turn to the Persian copular agreement pattern in Section 7.

#### 7. BACK TO PERSIAN

In the previous section we provided an analysis of NP2 agreement in Eastern Armenian copular clauses, which arises even in assumed identity contexts,

<sup>[26]</sup> Note that every set is a superset of itself.

<sup>[27]</sup> In these schema, we abstract away from movement of NP1 to Spec, TP to satisfy the EPP. This movement is modeled in Béjar (2003) as part of the mechanics of defective intervention effects (following Chomsky 2001). The minimal feature on NP1 halts the probe even though it cannot agree, and NP1 then displaces to Spec, TP (to satisfy the EPP). This displacement is what clears the search space, making NP2 agreement possible.

contrary to standard assumptions. The question now is how to deal with languages like Persian, which have NP2 agreement in specificational contexts but not in assumed identity ones. The crux of the problem is that 3rd person NP1s in Persian-type languages pattern in two ways: they are transparent with respect to Agree in specificational contexts, but visible to Agree in assumed identity contexts. The system proposed for Eastern Armenian predicts (incorrectly for Persian) that a 1st or 2nd person NP2 should control agreement when NP1 is 3rd person, regardless of the copular clause type. What is required is a system where NP1 fails to Agree only in specificational clauses. We will show that our solution for Eastern Armenian can be extended to Persian, with a modification of the probe structure.

The question is what special property does the NP1 in a specificational clause have or lack? Similar to what we proposed for Eastern Armenian, we posit here that the transparency of the Persian NP1 in specificational clauses arises because its feature structure is impoverished relative to the feature structure of the probe (the latter being what sets the criteria for successful Agree). In Eastern Armenian we related the asymmetry between the probe and the transparent 3rd person NP1 to the presence/absence of a [participant] feature. For Persian specificational clauses, we take the goal to have an impoverished feature structure due to the special structure of specificational subjects. One of the defining properties of specificational subjects is that they can only be construed as intensional expressions. They lack an extensional interpretation and cannot therefore be used deictically. We follow Romero (2005), Heycock (2012) and, most specifically, Den Dikken (2006b) in treating the intensional NP as a concealed CP, along the lines of (57).<sup>28</sup>

## (57) Structure for NP1 in specificational clauses



<sup>[28]</sup> Whether the concealed CP is a reduced relative clause (Den Dikken 2006) or a concealed question (Romero 2005, Heycock 2012) is an issue we abstract away from here.

The special structure of the DP in (57) has consequences for its feature specification. For a typical DP, we take D to be the locus of phi-features. We assume that the phi-features on D enter the derivation unvalued, and that they are valued via Agree with phi-features lower down in the nominal complex (e.g. # on NumP, gender on the root) (Adger & Harbour 2007, Danon 2011, Kucerova 2014) or that they are valued contextually by a deictic assignment function (Kucerova 2014). The D in (57) has a defective phi-bundle. This is because a full complement of phi-features would not be valued by phi-features lower in the structure – e.g. the phi-features on *murderer* in (57) – as they are contained inside a CP. Furthermore, given the strictly intensional interpretation available to the DP, there cannot be an assignment of phi-features to D via deixis. This is comparable to the unavailability of temporal deixis for defective T.

We model the defectiveness of D by positing an impoverished phi-structure. Rather than having a full complement of phi-features, we take the defective D to have just the minimal feature required to be recognized in the syntax as a nominal category. We call this feature [n].<sup>29</sup> We can now model the Persian pattern using the same kind of probe–goal mechanics that we saw above. The only difference is in the articulation of the probe. For Persian, we take the Probe on T to minimally involve  $[\_n]$  and  $[\_d]$ . The feature [n] on the intensional DP is thus insufficient to satisfy the more richly articulated probe. In (58) we show how these assumptions combine to derive the transparency of NP1 in the case of a Persian specificational clause. As in Eastern Armenian, we take the location of AGR in (58) to be on T for reasons that include a NOM–NOM case pattern and the impossibility of the accusative marker *-raa* on either NP1 or NP2. Note that the order of NP1 and NP2 in (58) is the output of the inversion step schematized in (6a) above.<sup>30</sup>

(i) Specificational context in Eastern Armenian

 AGR	 NP1	 NP2	
[_ <i>n</i> ]	[ <i>n</i> ]	[ <i>n</i> ]	
$[\_d]$		[d]	
[_part]		[part]	
		([speak	(er]

<sup>[29]</sup> We do not necessarily take [n] to represent a category feature, though this is a possibility we do not rule out. The main idea is that [n] is the minimal specification of a phi-bundle which can therefore potentially Match a phi-probe.

<sup>[30]</sup> We take the defective feature structure for the intensional NP1 in Persian to be a general characteristic of specificational subjects which we extend to Eastern Armenian as well. The derivation of NP2 agreement in Eastern Armenian specificational clauses falls out in the expected fashion, as shown in (i), where we have updated the feature structures to include the [n] feature:

(58) Specificational context in Persian

 AGR	NP1	NP2
[_ <i>n</i> ]	[ <i>n</i> ]	[ <i>n</i> ]
$[\_d]$		[d]
		([part])
		([speaker])

Let us now turn to the problem of Persian assumed identity contexts, where unlike Eastern Armenian, a 3rd person NP1 is not transparent. Crucially, an extensional NP1 does not have the structure in (57). It is a regular, non-defective DP, and will have canonical phi-features. As noted in Section 6, we take the minimal phi-structure of an extensional DP to include [n] and [d].<sup>31</sup> The result, as shown in (59), is that in Persian an extensional NP1 will always satisfy the probe thereby blocking NP2 agreement. In (59) this is schematized for a 3rd person NP1, however the same result would obtain if NP1 was a more richly articulated 1st or 2nd person.

(59) Assumed identity context in Persian

In this section, we have shown how the agreement facts in Persian copular clauses can be treated as a phi-sensitivity phenomenon, given the right probe structure. In doing so, we have unified the accounts for the Persian and Eastern Armenian agreement patterns and reduced their differences to variation in the probe structure. In Eastern Armenian, the probe is more finely articulated. It includes a [participant] feature, and as such overlooks a 3rd person NP1 regardless of whether it is extensional or intensional. In Persian, the probe is articulated only as far as the deictic feature [d], and consequently an extensional NP is sufficient for agreement – regardless of its person specification. This gave us the crucial difference in the agreement behaviour of the two languages in assumed identity contexts.<sup>32</sup>

<sup>[31]</sup> For personal pronouns and animate NPs additional features like [ $\pi$ ] may be warranted, but, as noted in footnote 25 above, they do not seem relevant to the patterns we are discussing here.

<sup>[32]</sup> Note that for Persian, as for Eastern Armenian (see footnote 23 above), number does not determine the outcome of agreement. It is parasitic on person agreement. We see this in (8c), (10d–f), (64a, b), (65a, b) and (66a, b). The crucial case is (66a), where we have singular NP1 agreement even though there is a plural NP2 in the clause. In this example, even though a plural NP is available in the search domain, it does not control agreement because the probe is only sensitive to person features which are satisfied by the higher singular NP.

In the above discussion, we have abstracted away from contexts involving two 3rd person NPs. This is the topic of the following section.

#### 8. COPULAR CLAUSES WITHOUT PARTICIPANTS

In this section we investigate the agreement patterns in Persian and Eastern Armenian copular clauses without a participant NP, i.e. copular clauses with two 3rd person NPs. In fact, copular clauses with two 3rd person NPs were the contexts originally noted in the literature on NP2 agreement (for example, Moro 1997 for Italian, Costa 2004 for Portuguese), where it was shown that, in specificational clauses with a singular NP1 and a plural NP2, agreement is with the plural NP2, as shown in the following famous example from Moro (Moro 1997: 28):

(60) la causa della rivolta furono/\*fu le foto del muro the cause of.the riot were/was the pictures of.the wall

We have seen above that Persian behaves like these languages showing NP2 agreement in specificational contexts involving a participant. The expectation is that the same pattern will extend to copular clauses with two 3rd person NPs. This is what is predicted by the system proposed in the previous section as well. The intensional NP1 will be transparent leading to agreement with the extensional NP2. The feature structure of a 3rd person NP2 is sufficient in our system to value the probe. This prediction is borne out, as shown in (61).<sup>33,34</sup>

- (61) (a) barande-ye olampiyaad-e fizik daneshju-haa-ye daaneshgaah-e winner-EZ olympiad-EZ physics student-PL-EZ university-EZ Sharif-Ø-an Sharif-be-3PL
  'The winners of the physics olympiad are the Sharif University students.'
  - (b) moshkel-e asli rahbar-aa-ye enghelaab-Ø-an problem-EZ main leader-PL-EZ revolution-be-3PL

'The main problem is the leaders of the revolution.'

More generally, in 3rd person contexts, our analysis predicts the attested outcomes. In predicational and assumed identity copular clauses, we should see agreement with the first 3rd person NP because it is extensional. The relevant examples are given below. In predicational clauses exemplified in (62), agreement is with the extensional plural NP1.

<sup>[33]</sup> It is not possible to have a plural NP1 and a singular NP2 in this context for pragmatic reasons.

<sup>[34]</sup> This pattern was also seen in (10f) above with an identificational example, which we have taken to be a subtype of specificational clauses.

- (62) (a) hame baraadar-haa-ye Ali doktor-Ø-an all brother-PL-EZ Ali doctor-be-3PL
   'All Ali's brothers are doctors.'
  - (b) daaneshjuhaa-(y)e Kamran tanhaa negaraani-Ø-an students-EZ Kamran only concern-be-3PL
     'Kamran's students are the only concern.'

In assumed identity (equative) contexts, where we can vary the number on NP1 and NP2, agreement is nevertheless with the first NP, singular in (63a) and plural in (63b).

- (63) (a) diruz, Sabah Kamnoosh Soroosh o Arsalan bud yesterday Sabah Kamnoosh Soroosh and Arsalan be.PST.3SG
   'Yesterday, Sabah was Kamnoosh, Soroosh and Arsalan.'
  - (b) diruz, Kamnoosh, Soroosh o Arsalan Sabah bud-an yesterday Kamnoosh, Soroosh and Arsalan Sabah be.PST-3PL
     'Yesterday, Kamnoosh, Soroosh and Arsalan were Sabah.'

The context for these examples is the game of charades discussed above and exemplified for Persian in (14). In (63a), we can take Sabah to have acted out the three individuals in NP2 and in (63b), we can take the three individuals in NP1 to have all acted out Sabah in a game of charades played the previous day.

So far we have noted that agreement in Persian copular clauses with two 3rd person NPs is exactly as predicted by the system developed in Section 7. Agreement is always with an extensional NP and if there are two extensional NPs, with the first one. The question becomes more interesting when we look at a language like Eastern Armenian. Recall from the discussion in Section 6 that in Eastern Armenian agreement is consistently with the participant NP when there is one. We attributed this agreement pattern to the probe having the feature [participant] articulated in its feature structure. When both NPs are participants, NP1 controls agreement, as expected. The question we are raising now is what happens when there is no participant NP in the copular clause and the search for a matching participant goal fails. An examination of the relevant facts in Eastern Armenian reveals that in the absence of a participant NP, agreement is with the highest extensional NP. In other words, when both NPs are 3rd person, Eastern Armenian behaves exactly like Persian. The Eastern Armenian facts are given in (64)-(66). In (64), we have examples of specificational copular clauses, where agreement is with the plural extensional NP2.

(64) (a) fizik-i olimpiad-i tanoq-ə Sharif hamalsaran-i Physics-GEN Olympiad-GEN winner-SP Sharif university-GEN ašakert-ner-ən en student-PL-SP are
'The winners of the physics olympiad are the Sharif University students.'  (b) iskakan problem-ə heqapoxutyan metz-er-ən en main problem-SP revolution(GEN) chief-PL-SP are
 'The main problem is the leaders of the revolution.'

Examples of predicational clauses are given in (65) where agreement is with the extensional NP1.

- (65) (a) Ara-i bolor yexpayr-ner-ə doktor en Ara-GEN all brother-PL-SP doctor are 'All Ali's brothers are doctors.'
  - (b) Kamran-i ašakert-ner-ə miak hoks-ən en Kamran-GEN student-PL-SP only concern-SP are 'Kamran's students are the only concern.'

In (66), where both NPs are extensional, agreement is with the higher NP1.

- (66) (a) erek Sabah-Ə Kamnoosh-(Ə) Soroosh-(Ə) yev Arsalan-Ən yesterday Sabah-SP Kamnoosh-SP Soroosh-SP and Arsalan-SP er was
  'Yesterday, Sabah was Kamnoosh, Soroosh and Arsalan.'
  - (b) erek Kamnoosh Soroosh u Arsalan Sabah( on) ein yesterday Kamnoosh Soroosh and Arsalan-SP Sabah-SP were
     'Yesterday, Kamnoosh, Soroosh and Arsalan were Sabah.'

The question is how we can account for the above facts given the articulated [participant] probe we proposed in Section 6. We seem to need a mechanism which turns the Eastern Armenian probe into the Persian probe when the first cycle of Agree fails, i.e. no [participant] NP is found to match the articulated feature structure. A mechanism of this kind has been argued for in the literature on agreement. Béjar (2003) argues that a failed probe remains active in the derivation (see also Rezac 2003, Béjar & Rezac 2009) and proposes that the failure of the initial cycle of Agree results in a reduction of the probe's feature structure. This structural change, referred to here as Probe Reduction and formulated in (67), augments the set of possible controllers by reducing the criteria for Match.

(67) Probe Reduction

Failure to Agree impoverishes the feature structure of a probe.

We now extend the notion of Probe Reduction to the context of agreement in Eastern Armenian copular clauses with two 3rd person NPs. When the first cycle of Agree fails in the absence of participant noun phrases, the feature structure of the probe is impoverished from the bottom, eliminating the [participant] feature and resulting in a probe structure identical to that of Persian. This mechanism and how it leads to agreement with the extensional NP2 in a specificational clause is shown schematically in (68), where shading indicates probe reduction.

(68) (a) Failure to Agree with 3rd person NP1 in first cycle

$$... \text{ AGR } ... \text{ NP1(3) } ... \text{ NP2(3) } ... \\ [\_n] [n] [n] [n] \\ [\_d] [d] [d] \\ [\_part]$$

(b) Probe Reduction and Agreement with NP2 in second cycle

The predicational and assumed identity (equative) cases can be accounted for in a similar fashion. The only difference has to do with which NP is the closest match for this reduced feature structure. In these cases, it is the higher NP1 that controls agreement, as expected (compare (59)).<sup>35</sup>

In this section we have investigated the agreement pattern of copular clauses with two 3rd person NPs. We showed for Persian that agreement alternates between NP1 and NP2 as predicted by the probe structure proposed in Section 7. For Eastern Armenian we addressed the derivational consequences of having an articulated [participant] probe in clauses with two 3rd person NPs. We proposed that a derivational mechanism (Probe Reduction) makes the highest extensional NP a Matching goal. This correctly predicts NP2 agreement in Eastern Armenian 3rd person specificational clauses, but NP1 agreement in predicational and assumed identity (equative) contexts.

## 9. CONCLUSION

In this paper we have investigated the availability of non-canonical NP2 agreement in copular clauses in two understudied languages, Persian and Eastern Armenian. We began by showing that this pattern does not arise from ordinary subject agreement where NP2 is the structural subject, with NP1 fronted. We further showed that a strict correlation between NP2 agreement and the syntax of inverted specificational clauses is not adequate to capture a full range of cross-linguistic patterns. We have argued instead that the derivational conditions

<sup>[35]</sup> In Béjar & Kahnemuyipour (2014), we provide further support for the probe reduction mechanism using data from Persian copular clauses embedded under the modal verb *tavaanestan* 'can'. We show that an intensional NP, which is transparent to agreement in the copular domain, becomes visible and controls agreement when raised to the modal domain. We argue that this happens in the higher modal domain as a result of the failure of the search for an extensional NP and the reduction of the probe structure to one which can be satisfied by an intensional NP. Such cases provide further evidence for the phi-sensitivity account and the implementation of the probe reduction mechanism.

giving rise to NP2 agreement reduce fundamentally to properties of the probegoal system, and can be treated as a person-sensitivity familiar from a broader range of contexts. Whereas person-sensitivities are typically framed in terms of more salient phi-features relating to person and animacy, the data in this paper suggest that in fact the relevant features for some languages may be even higher in the phi-feature entailment structure (our [d] and [n]) (see also Cowper & Hall 2002). Different articulations of the probe, cross-linguistically, yield contrasting agreement outcomes, leading to the different types of systems discussed here. While we have focused on Persian and Eastern Armenian in this paper, this set of proposals plausibly extends to languages with similar properties. The Persian pattern is instantiated in other languages (e.g. Romance, Germanic and Turkic). The Eastern Armenian pattern seems more unusual but may turn out to be less so if more languages are thoroughly investigated. These systems do not exhaust the attested range of variation, for instance a prominent third pattern is uniform NP1 agreement even in specificational contexts, as found in English (1a) above and French. We leave to future research the possibility that the pattern found in these languages might also be analyzed as a reflex of differential probe articulation (e.g. minimal [n]) or Probe Reduction.

#### REFERENCES

- Adger, David & Daniel Harbour. 2007. The syntax and syncretisms of the Person Case Constraint. *Syntax* 10, 2–37.
- Adger, David & Gillian Ramchand. 2003. Predication and equation. Linguistic Inquiry 34.3, 325–359.

Barlow, Michael & Charles A. Ferguson. 1988. Agreement in natural language. Stanford, CA: CSLI.

Béjar, Susana. 2003. Phi-syntax: A theory of agreement. Ph.D. dissertation, University of Toronto.

- Béjar, Susana. 2008. Conditions on Phi-Agree. In Daniel Harbour, David Adger & Susana Béjar (eds.), Phi theory: Phi-features across interfaces and modules, 130–154. Oxford: Oxford University Press.
- Béjar, Susana & Arsalan Kahnemuyipour. 2014. Agree and the (in)visibility of intensional NPs. Presented at the GETEGRA Agreement Workshop, Recife, Brazil.

Béjar, Susana & Milan Rezac. 2009. Cyclic Agree. Linguistic Inquiry 40.1, 35-73.

- Bobaljik, Jonathan David & Phil Branigan. 2006. Eccentric agreement and multiple case-checking. In Alana Johns, Diane Massam & Juvenal Ndayiragije (eds.), *Ergativity: Emerging issues* (Studies in Natural Language and Linguistic Theory 65), 47–77. Dordrecht: Springer.
- Bondaruk, Anna. 2013. Copular clauses in English and Polish: Structure, derivation and interpretation. Lublin: Wydawnictwo KUL.
- Bowers, John. 1993. The syntax of predication. Linguistic Inquiry 24.4, 591-656.
- Chomsky, Noam. 2000. Minimalist inquiries: The framework. In Roger Martin, David Michaels & Juan Uriagereka (eds.), *Step by step: Essays on Minimalist syntax in honor of Howard Lasnik*, 89–156. Cambridge, MA: MIT Press.
- Chomsky, Noam. 2001. Derivation by phase. In Michael Kenstowicz (ed.), Ken Hale: A life in language, 1–52. Cambridge, MA: MIT Press.
- Comrie, Bernard. 1984. Some formal properties of focus in Modern Eastern Armenian. Annual of Armenian Linguistics 5, 1–21.
- Comrie, Bernard. 2003. When agreement gets trigger happy. *Transactions of the Philological Society* 101.2, 313–337.
- Costa, João. 2004. Subjects in Spec, vP: Locality and agree. In Ana Castro, Marcelo Ferreira, Valentine Hacquard & Andres P. Salanova (eds.), *Romance, Op. 47: Collected papers on Romance syntax* (MIT Working Papers in Linguistics 47), 1–15. Cambridge, MA: MIT.

- Cowper, Elizabeth & Daniel Currie Hall. 2002. The syntactic manifestation of nominal feature geometry. In Sophie Burelle & Stanca Somesfalean (eds.), *Proceedings of the 2002 Annual Conference of the Canadian Linguistic Association*, 55–66. Montreal: Université du Québec à Montréal.
- Danon, Gabi. 2011. Agreement and DP-internal feature distribution. Syntax 14, 297-317.
- Declerck, Renaat. 1988. *Studies on copular sentences, clefts and pseudo-clefts*. Leuven: Leuven University Press and Foris Publications.
- Den Dikken, Marcel. 1997. Appraising *The raising of predicates*. Review article on *The raising of predicates: Predicative noun phrases and the theory of clause structure* by Andrea Moro. *Linguistische Berichte* 174, 246–263.
- Den Dikken, Marcel. 2006a. Specificational copular sentences and pseudoclefts. In Martin Everaert & Henk van Riemsdijk (eds.), *The Blackwell companion to syntax*, vol. IV, 292–409. Oxford: Wiley-Blackwell.
- Den Dikken, Marcel. 2006b. Relators and linkers. Cambridge, MA: MIT Press.
- Dixon, R. M. W. 2002. Copular clauses in Australian languages: A typological perspective. Anthropological Linguistics 44, 1–36.
- Doron, Edit. 1988. The semantics of predicate nominal. Linguistics 26.2, 281-301.
- Geist, Ljudmila. 2003. Predication and equation in copular clauses: The syntax-semantics interface. In Ileana Comorovsky & Klaus von Heusinger (eds.), *Existence: Semantics and syntax*, 79–105. Dordrecht: Springer.
- Hale, Kenneth. 2001. Eccentric agreement. In Pablo Albizu & Beatriz Fernández (eds.), On Case and agreement, 15–48. Bilbao: Euskal Herriko Unibertsitatea.
- Halle, Morris. 1997. Distributed morphology: Impoverishment and Fission. In Benjamin Bruening, Yoonjung Kang & Martha McGinnis (eds.), *PF: Papers at the interface* (MIT Working Papers in Linguistics 30), 425–450. Cambridge, MA: MIT.
- Harley, Heidi & Elizabeth Ritter. 2002. Person and number in pronouns: A feature-geometric analysis. *Language* 78.3, 482–526.
- Heggie, Lorie. 1988. The syntax of copular structures. Ph.D. dissertation, University of Southern California.
- Heller, Daphna. 2005. *Identity and information: Semantic and pragmatic aspects of specificational sentences*. Ph.D. dissertation, Rutgers University.
- Heller, Daphna & Lindsey Wolter. 2008. That is Rosa: Identificational sentences and intensional predication. In Atle Groen (ed.), *Proceedings of Sinn und Bedeutung* 12, 226–240. Oslo: Department of Literature, Area Studies and European Languages, University of Oslo. [http://www.hf.uio.no/ilos/f orskning/konferanser/SuB12/proceedings/, accessed 26 April 2016]
- Heycock, Caroline. 1994. Layers of predication: The non-lexical syntax of clauses. New York: Garland.
- Heycock, Caroline. 2010. Variability and variation in agreement in copular clauses: Evidence from Faroese. Presented at the 25th Comparative Germanic Syntax Workshop (CGSW 25), June 2010, University of Tromsø, Norway. [http://www.lel.ed.ac.uk/~heycock/papers/CGSW25.pdf, accessed 26 April 2016]
- Heycock, Caroline. 2012. Specification, equation and agreement in copular sentences. *Canadian Journal of Linguistics* 57.2, 209–240.
- Heycock, Caroline & Anthony Kroch. 1998. Inversion and equation in copular sentences. In Artemis Alexaidou, Nanna Fuhrhop, Ursula Kleinhenz & Paul Law (eds.), *ZAS Papers in Linguistics* 10, 71–87. Berlin: Zentrum für Allgemeine Sprachwissenschaft.
- Heycock, Caroline & Anthony Kroch. 1999. Pseudocleft connectedness: Implications for the LF interface level. *Linguistic Inquiry* 30.3, 365–397.
- Higgins, Roger F. 1973. The pseudo-cleft construction in English. Ph.D. dissertation, MIT.
- Higgins, Roger F. 1979. The pseudo-cleft construction in English. New York: Garland.
- Kahnemuyipour, Arsalan. 2001. On *wh*-questions in Persian. *Canadian Journal of Linguistics* 46.1–2, 41–61.
- Kahnemuyipour, Arsalan. 2009. The syntax of sentential stress. Oxford: Oxford University Press.
- Kahnemuyipour, Arsalan. 2014. Revisiting the Persian Ezafe construction: A roll-up movement analysis. *Lingua* 150, 1–24.
- Kahnemuyipour, Arsalan & Karine Megerdoomian. 2011. Second position clitics in the vP phase: The case of the Armenian auxiliary. *Linguistic Inquiry* 42.1, 152–162.
- Kahnemuyipour, Arsalan & Karine Megerdoomian. 2017. On the positional distribution of an Armenian auxiliary: Second-position clisis, focus, and phases. *Syntax* 20.1, 77–97.

- Kucerova, Ivona. 2014. Defective Agree Chains in Italian nominal inflection. Presented at the 44th Linguistic Symposium on Romance Languages (LSRL), Western University, London, ON, Canada.
- Laka, Itziar. 1993. Unergatives that assign ergative, unaccusatives that assign accusative. In Jonathan D. Bobaljik & Chris Collins (eds.), *Papers on Case and agreement I* (MIT Working Papers in Linguistics 18), 149–172. Cambridge, MA: MIT.
- Macaulay, Monica. 1992. Inverse marking in Karuk: The function of the suffix *-ap. International Journal of American Linguistics* 58, 182–201.
- Mikkelsen, Line. 2005. Copular clauses: Specification, predication and equation. Amsterdam & Philadelphia, PA: John Benjamins.
- Moltmann, Friederike. 2010. Identificational sentences and the objects of direct perception. Ms., IHPST (Paris1/ENS/CNRS). [http://semantics.univ-paris1.fr/pdf/present-pronouns-2010-pub 1-2.pdf, accessed 16 December 2016]
- Moro, Andrea. 1988. Per una teoria unificata delle frasi copulari [Towards a unified theory of copular clauses]. *Rivista di Grammatica Generativa* 13.8, 1–110.
- Moro, Andrea. 1997. The raising of predicates: Predicative noun phrases and the theory of clause structure. Cambridge: Cambridge University Press.
- Moro, Andrea. 2006. Copular sentences. In Martin Everaert & Henk van Riemsdijk (eds.), Blackwell companion to syntax, vol. II, 1–23. Oxford: Wiley-Blackwell.
- Moro, Andrea. 2013. The equilibrium of human syntax: Symmetries in the brain. New York: Routledge.
- Nichols, Lynn. 2001. The syntactic basis of referential hierarchy phenomena: Clues from languages with and without morphological case. *Lingua* 111, 515–537.
- Partee, Barbara. 1986. Ambiguous pseudoclefts with unambiguous be. Proceedings of the Sixteenth Annual Meeting of the North East Linguistic Society (NELS 16), 354–366. Amherst, MA: GLSA.
- Pereltsvaig, Asya. 2007. Copular sentences in Russian: A theory of intra-clausal relations. New York: Springer.
- Polinsky, Maria. 2003. Non-canonical agreement is canonical. *Transactions of the Philological Society* 101.2, 279–312.
- Pustet, Regina. 2003. Copulas: Universals in the categorization of the lexicon. Oxford: Oxford University Press.
- Rapoport, Tova. 1987. Copular, nominal and small clauses: A study of Israeli Hebrew. Ph.D. dissertation, MIT.
- Rezac, Milan. 2003. The fine structure of cyclic Agree. Syntax 6, 156–182.
- Rizzi, Luigi. 1982. Issues in Italian syntax. Dordrecht: Foris.
- Rizzi, Luigi. 1990. Relativized Minimality. Cambridge, MA: MIT Press.
- Romero, Maribel. 2005. Concealed questions and specificational subjects. *Linguistics and Philosophy* 28, 687–737.
- Rothstein, Susan. 1983. The syntactic form of predication. Ph.D. dissertation, MIT.
- Rothstein, Susan. 1995. Small clauses and copula constructions. In Anna Cardinaletti & Maria Teresa Guasti (eds.), *Small clauses* (Syntax and Semantics 28), 27–48. San Diego, CA: Academic Press.
- Rothstein, Susan. 2001. Predicates and their subjects. Dordrecht: Kluwer.
- Sharvit, Yael. 1999. Connectivity in specificational sentences. *Natural Language Semantics* 7.3, 299–339.
- Sigurðsson, Halldór A. 1996. Icelandic finite verb agreement. *Working Papers in Scandinavian Syntax* 57, 1–46.
- Sigurðsson, Halldór A. 2006. The Nom/Acc alternation in Germanic. In Jutta M. Hartmann & László Molnárfi (eds.), Comparative studies in Germanic syntax, 13–50. Amsterdam: John Benjamins.
- Silverstein, Michael. 1976. Hierarchy of features and ergativity. In R. M. W. Dixon (ed.), Grammatical categories in Australian languages, 112–171. Canberra: Australian Institute of Aboriginal Studies. Stassen, Leon. 1997. Intransitive predication. Oxford: Oxford University Press.
- Tamrazian, Arminé. 1994. The syntax of Armenian: Chains and the auxiliary. Ph.D. dissertation,
- University College London.
- Taraldsen, Knut T. 1995. On agreement and nominative objects in Icelandic. In Hubert Haider, Susan Olsen & Sten Vikner (eds.), Studies in comparative Germanic syntax, 307–327. Dordrecht: Kluwer.
- Tragut, Jasmine. 2009. Armenian: Modern Eastern Armenian. Amsterdam: John Benjamins.
- Zwicky, Arnold M. 1977. Hierarchies of person. In Woodford A. Beach, Samuel E. Fox & Shulamith Philosoph (eds.), *The Thirteenth Annual Meeting of the Chicago Linguistic Society* (CLS 13), 714–733. Chicago, IL: University of Chicago, Chicago Linguistic Society.

#### NON-CANONICAL AGREEMENT IN COPULAR CLAUSES

#### Authors' addresses: (Béjar)

Department of Linguistics, University of Toronto, 100 St. George Street, Toronto, ON M5S 3G3, Canada sbejar@chass.utoronto.ca

(Kahnemuyipour)

Department of Language Studies, University of Toronto Mississauga, 3359 Mississauga Road, Mississauga, ON L5L 1C6, Canada a.kahnemuyipour@utoronto.ca